ECONSTOR Make Your Publications Visible.

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Fitzenberger, Bernd; Fuchs, Benjamin

Working Paper

The residency discount for rents in Germany and the Tenancy Law Reform Act 2001: Evidence from quantile regressions

SOEPpapers on Multidisciplinary Panel Data Research, No. 822

Provided in Cooperation with: German Institute for Economic Research (DIW Berlin)

Suggested Citation: Fitzenberger, Bernd; Fuchs, Benjamin (2016) : The residency discount for rents in Germany and the Tenancy Law Reform Act 2001: Evidence from quantile regressions, SOEPpapers on Multidisciplinary Panel Data Research, No. 822, Deutsches Institut für Wirtschaftsforschung (DIW), Berlin

This Version is available at: https://hdl.handle.net/10419/130575

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



SOEPpapers

on Multidisciplinary Panel Data Research

SOEP - The German Socio-Economic Panel study at DIW Berlin

822-2016

522¹⁰

The Residency Discount for Rents in Germany and the Tenancy Law Reform Act 2001: Evidence from Quantile Regressions

Bernd Fitzenberger and Benjamin Fuchs



SOEPpapers on Multidisciplinary Panel Data Research at DIW Berlin

This series presents research findings based either directly on data from the German Socio-Economic Panel study (SOEP) or using SOEP data as part of an internationally comparable data set (e.g. CNEF, ECHP, LIS, LWS, CHER/PACO). SOEP is a truly multidisciplinary household panel study covering a wide range of social and behavioral sciences: economics, sociology, psychology, survey methodology, econometrics and applied statistics, educational science, political science, public health, behavioral genetics, demography, geography, and sport science.

The decision to publish a submission in SOEPpapers is made by a board of editors chosen by the DIW Berlin to represent the wide range of disciplines covered by SOEP. There is no external referee process and papers are either accepted or rejected without revision. Papers appear in this series as works in progress and may also appear elsewhere. They often represent preliminary studies and are circulated to encourage discussion. Citation of such a paper should account for its provisional character. A revised version may be requested from the author directly.

Any opinions expressed in this series are those of the author(s) and not those of DIW Berlin. Research disseminated by DIW Berlin may include views on public policy issues, but the institute itself takes no institutional policy positions.

The SOEPpapers are available at http://www.diw.de/soeppapers

Editors:

Jan **Goebel** (Spatial Economics) Martin **Kroh** (Political Science, Survey Methodology) Carsten **Schröder** (Public Economics) Jürgen **Schupp** (Sociology)

Conchita **D'Ambrosio** (Public Economics, DIW Research Fellow) Denis **Gerstorf** (Psychology, DIW Research Director) Elke **Holst** (Gender Studies, DIW Research Director) Frauke **Kreuter** (Survey Methodology, DIW Research Fellow) Frieder R. **Lang** (Psychology, DIW Research Fellow) Jörg-Peter **Schräpler** (Survey Methodology, DIW Research Fellow) Thomas **Siedler** (Empirical Economics) C. Katharina **Spieß** (Education and Family Economics) Gert G. **Wagner** (Social Sciences)

ISSN: 1864-6689 (online)

German Socio-Economic Panel (SOEP) DIW Berlin Mohrenstrasse 58 10117 Berlin, Germany

Contact: Uta Rahmann | soeppapers@diw.de



The Residency Discount for Rents in Germany and the Tenancy Law Reform Act 2001: Evidence from Quantile Regressions^{*}

BERND FITZENBERGER^{\ddagger} and BENJAMIN FUCHS^{\ddagger}

 $^{\natural}$ Humboldt-University Berlin, IZA, CESifo, IFS, ROA, ZEW, bernd.fitzenberger@wiwi.huberlin.de

[#] University of Hohenheim, benjamin.fuchs@uni-hohenheim.de

February 2016

Abstract: Most countries show a residency discount in rents for sitting tenants. In the wake of strong rent increases and housing shortages, Germany implemented a reform in 2001 to curtail rent increases. Based on linked housing-tenant data for Germany, this paper estimates panel OLS and quantile regressions of rents within tenancies. The results show that rents deflated by the CPI increase strongly from 1984 until the reform in 2001, and there is a reversal in the trend afterwards. Before the reform, there is a significant residence discount which decreases in absolute value with tenure. The reform reduces rents, in particular for expensive apartments and for new leases. There is no residency discount after the reform.

Keywords: linked housing-tenant data, rent regression, length of residency discount, rent control, quantile regression

JEL: R 31, C 21, C 23

Corresponding Author: Bernd Fitzenberger, Humboldt-University Berlin, School of Business and Economics, Spandauer Strasse 1, 10099 Berlin, Germany

^{*}The paper benefited from many helpful comments by two anonymous referees as well as by seminar and conference participants in Berlin, Düsseldorf, Essen, and Freiburg. This paper uses data from the German Socio-Economic Panel Study that has been made available by the German Institute for Economic Research (DIW), Berlin.

1 Introduction

In comparison to other European countries, a fairly large share of the population in Germany lives in rental housing (Eurostat 2013). At the same time, the large private rental housing market in Germany shows a higher level of regulation through rent control and tenant protection against eviction than the OECD average (Johannsson 2011). In contrast to the current situation, Germany was characterized until the 1990s by fairly liberal rent laws (Börsch-Supan 1994; Hubert 1998).¹ In the wake of housing shortages and strong rent increases in the 1990s, there was a change towards more regulation in order to protect sitting tenants. There are less restrictions on rent increases for new leases and these restrictions remained basically unchanged during the 1990s and 2000s.

The rental housing market in most countries shows a residency discount for sitting tenants. With stronger rent control for sitting tenants and tenant protection against eviction, one would expect a stronger residency discount for sitting tenants if rents for new leases can adjust to market conditions (Börsch-Supan 1994; Hubert 1995, 1998). Thus, rent control is likely to protect sitting tenants more than tenants in new leases. This paper analyzes the impact of the Tenancy Law Reform Act implemented in 2001 on both the level of rents and the residency discount. Based on linked housing-tenant data from 1984 to 2011, we estimate panel OLS and quantile regressions of rents within tenancies.

As an application of hedonic price models (Court 1939, Rosen 1974), a large body of empirical research examines the relationship between characteristics of rental units and average rents (see e.g. Barnett 1979, Follain and Malpezzi 1980, Guasch and Marshal 1987 for the U.S. or Hoffmann and Kurz 2002 or Bischoff and Maennig 2011 for Germany). Rents are regressed on characteristics of rental units and the coefficients are interpreted as the marginal prices for these characteristics. In an unregulated rental market, the price effects reflect market conditions which are the result of supply and demand. In a regulated market, which is the case in our paper, the estimated price effects are also affected by the institutional constraints. Furthermore, not all price relevant characteristics are observed in rent data and the

¹Currently (in 2015), the regulation of the private rental housing market in Germany is further strengthened through stronger rent control ('Mietpreisbremse') both for new leases and for sitting tenants and through the requirement that the person, who engages a real estate agent to find a tenant/an apartment, has to pay for the service her-/himself. Thus, landlords cannot add the service fee for an agent they engaged to the rents the tenants have to pay.

distribution of unobserved characteristics may change with observed characteristics. Thus, observed characteristics may have different implicit prices along the rent distribution. The latter aspects motivate the estimation of quantile regressions which go beyond the estimation of the effects on average rents.

Hedonic price regressions are often augmented by covariates which are not per se characteristics of the rental units and which may reflect frictions in the rental market. Regulations such as rent control and tenant protection against eviction have an impact on rents and the effects may differ by the level of rents. For instance, one would expect that rent control should reduce the rents for expensive apartments more strongly than for cheap apartments. Similarly, rent control for sitting tenants should increase the length of residency discount, i.e. the reduction in rents as elapsed tenure of the tenant grows, when rents for new leases are less regulated (as it is the case in Germany). This effect may be strongest for expensive rental units, which further motivates the estimation of quantile regressions for rents.

The rent control and the protection of tenants against eviction were strengthened in light of the rent increases observed in West Germany during the 1990s. The Tenancy Law Reform Act, implemented since September 1, 2001, covers all tenancies starting on this date or later. Its most substantial changes involve a cap on the maximum rent increases by landlords for sitting tenants and a reduction of the minimum notice time until termination of a tenancy by the tenant to three months, while keeping the protection of the tenant against eviction unchanged. Before the reform, landlords could increase rents by 30% within three years. After the reform, the maximum is reduced to 20%.² Our empirical analysis investigates the impact of the reform on rents. Specifically, we investigate the observed change in the length of residency discount because the reform did not change regulations of the rents for new leases.

Our empirical analysis uses linked housing-tenant data from the German Socio-Economic Panel (SOEP) for the time period 1984 to 2011. These data offer the unique advantage of a large and representative panel data set of tenancies. The panel structure allows us to control for unobservable time-invariant characteristics of a tenancy. We restrict the analysis to West German households and to observations with elapsed tenure up to 10 years.

²The actual upper limit of rent increases, however, was and still is given by the average local rent index (published by local authorities) reflecting the average rent of comparable rental units in the neighborhood ('ortsübliche Vergleichsmiete'). A higher rent increase is possible to compensate for the costs of modernizing the rental unit or for an increase in the running costs.

Our empirical results show that rents deflated by the CPI increase strongly from 1984 until the reform in 2001, and there is a reversal in the trend afterwards. Before the reform, we find a significant residence discount in the rent level which decreases in absolute value with tenure and which is stronger at the top than at the bottom of the rent distribution. The reform reduces rents, in particular for expensive apartments and for new leases. The reform eliminates the residency discount and the fixed effects estimates reveal that after the reform rents in fact grow with tenure, especially for more expensive apartments. The evidence is consistent with expensive tenancies being likely to end earlier and tenants in new leases benefitting strongly from a reduction in rents induced by the reform. Our evidence suggests that the reform was successful in curtailing rent increases for new leases and for expensive apartments early in a tenancy.

The remainder of this paper is organized as follows. Section 2 discusses the background of the analysis and reviews the existing literature. Section 3 describes the data used. Section 4 introduces the econometric approach. Section 5 provides descriptive statistics and discusses the regression results. The final section involves our concluding remarks. An additional online appendix contains all the results, which in this paper are referred to as being available upon request.³

2 Background

We first discuss economic and institutional aspects of rental housing in Germany. Then, we provide a selective literature review and develop some hypotheses as the basis of our empirical analysis.

2.1 Private Rental Housing in Germany

Evidence provided by Eurostat (2013)⁴ shows that in 2013 about 47% of the German population live in rental housing, which is a high share in international comparison. Only in Switzerland, this share is even higher while especially in Eastern and Southern Europe owner-occupied housing is much more common. The causes for this difference are manifold. Differences in attitudes towards home-ownership may play a role. While in Germany home ownership is rather viewed as a long-term invest-

³See www.wiwi.hu-berlin.de/de/professuren/quantitativ/oe/research/publications. ⁴Online data code: ilc_lvho02.

ment and rental housing seems better suited for temporary housing needs, it is more common in Anglo-Saxon countries to buy and sell residential houses depending on the own current economic situation and needs (Börsch-Supan 1994). In addition to differences in the flexibility to buy and sell residential houses, preferential tax treatment may shape housing demand. In contrast to a number of other European countries, there is no preferential tax treatment of owner-occupied housing in Germany regarding the difference between the after-tax and the pre-tax interest rate of mortgage loans, which is a likely reason for the high share of rental housing in Germany (CESifo 2005).⁵

While Germany is characterized by rather liberal rental laws up to the early 1990s, a shift takes place since then towards much more "tenant friendly" rental laws (Hubert 1998, Börsch-Supan 1994). In comparison to other OECD countries, the private rented housing sector in Germany is strongly regulated curtailing the flexibility of landlords. Figures 1 and 2, as taken from Johansson (2011), show that Germany as of 2010 shows a very high level of rent control and an above average level of tenant protection in international comparison. Despite the tighter regulations, rents increase strongly during the 1990s in most parts of Germany, especially in the metropolitan areas. Hubert (1998) views the rising regulation in the 1990s as a response to an acute housing shortage after German unification in order to avoid strong rent increases for incumbent tenants. Because rent increases for new leases are less regulated (Börsch-Supan 1994), strong rent control for sitting tenants is complemented by tenant protection against eviction by the landlord motivated by the goal of increasing the rent (henceforth denoted as economic eviction).

— Insert figure 1 here. —

— Insert figure 2 here. —

As a further policy response to protect tenants, the Tenancy Law Reform Act^6 in 2001 reduces the maximum rent increases for sitting tenants from 30% to 20% over the course of three years. The new cap applies only, if the planned increase in rents does not exceed the average rent of comparable units in the neighborhood

⁵The current situation differs strongly from the rather favorable taxation of owner-occupied housing during the 1970s and 1980s in West Germany, see Hubert (1998) and Börsch-Supan (2004).

⁶Gesetz zur Neugliederung, Vereinfachung und Reform des Mietrechts (Mietrechtsreform), see Bundestag (2000) for a discussion of the objectives of the reform.

(ortsübliche Vergleichsmiete) as measured by the local rent index excluding rent increases due to modernization or growing running costs included in rents (see §558(3), §559, and §560 German Civil Code [BGB] for further details). The local rent index is intended to provide both tenants and landlords an indication of market conditions and to allow tenants to identify particularly high rents.⁷ A local rent index typically provides for comparable apartments the average rent and the interval around the average, covering the two-third span ranging from the one sixth (17%) quantile to the five sixth (83%) quantile. The local rent index serves as a monitoring instrument for rent control in Germany.

Furthermore, the reform involves a reduction of the minimum notice time until termination of a tenancy by the tenant to three months, while it keeps the protection of the tenant against eviction by the landlord unchanged. Before the reform, the notice time depends upon the length of the tenancy, with a minimum of three and a maximum of twelve months. The reform initially only applies to tenancies that started on September 1, 2001 or later. As an extension of the reform in 2005, the notice time of three months also applies to tenancies which has started before September 1, 2001.⁸ One objective of our paper is to estimate the statistical association between the 2001 reform and the length of residency discount over the distribution of rents.

2.2 Literature Review and Hypotheses

There exists a sizeable literature in economics on the effect of rent control on the rental market. Eekhoff (1981), Börsch-Supan (1986), and Schwager (1994) provide a theoretical discussion of the welfare implications of a reform in Germany in 1975, which strengthens the protection of tenants against economic eviction and which regulates maximum rent increases for sitting tenants.⁹ There is stronger rent control for sitting tenants compared to rent setting for new leases in order to protect sitting tenants. The predicted welfare effects are ambiguous. On the one hand, there are negative welfare effects because rent control reduces efficiency and flexibility in the rental market, thus resulting in deviations from market equilibrium and from

⁷Based on a sample from the local rental market, the local rent index (Mietspiegel) reports the average rents and the dispersion of rents for comparable apartments/housing units.

 $^{^8 {\}rm See}$ Klarstellungsgesetz 2005 and Art. 229 $\S 3$ Abs. 10 EGBGB - Einführungsgesetz BGB - for further information.

⁹"Law for the Protection of Tenants from Arbitrary Eviction" [Zweites Wohnraumkündigungsschutzgesetz (2. WKSchG)] The law is a predecessor of the 2001 Tenancy Law Reform Act.

the law-of-one-price for a good with the same characteristics. It may reduce the supply of rental units. On the other hand, there could be positive welfare effects if landlords value the curtailment of their property rights less than tenants value the benefits of a cap on rent increases in a dynamic perspective (Börsch-Supan 1994, Hubert 1995). Furthermore, rent control for sitting tenants is likely to imply a front loaded rent payment schedule, where landlords would ask for higher rents at the beginning of a tenancy to compensate for the stronger rent control during tenancy. This could result in rental payments that decrease with the length of residency for sitting tenants relative to rents for new leases of comparable apartments. Further implications of these considerations are that the rent for a given apartment increases more strongly than the market rents for comparable apartments when a new lease starts, and therefore landlords have an incentive to evict sitting tenants in order to realize a rent increase with a new lease. Because of the latter incentive, a stronger rent control for sitting tenants is typically complemented with a stronger protection of sitting tenants against economic eviction.

Before reviewing some empirical results on the length of residency discount, let us discuss some pertinent theoretical aspects in a bit more detail. A large part of the U.S. literature (e.g. Guasch and Marshall 1987) argues that the length of residency discount can be explained by the survival of good matches of landlords and tenants in the presence of turnover costs for both sides. Providing a somewhat different perspective, Barker (2003) considers the relationship between turnover costs and the level of price discrimination between new leases and long-term tenancies. Landlords of apartments with low turnover costs are more likely to raise rents for sitting tenants. Furthermore, tenants in new leases could obtain a discount because of a lower demand elasticity of long-term tenants or because of the higher mobility costs of the latter. Thus, it is an open empirical question as to whether a length of residency discount exists.

Relating the length of residency discount to regulation, Hubert (1995) discusses a possible justification for a regulation, which protects tenants against arbitrary eviction, based on efficiency grounds. The argument relies on adverse selection operating in the presence of asymmetric information about tenants. If tenants differ in the 'service costs' to be paid for by the landlords and landlords offer rental contracts with different lengths, then tenants with low service costs would select into shorter tenancies because they can show after a while that they are good (\equiv low service cost) tenants. Furthermore, longer tenancies would rather involve tenants with high ser-

vice costs. Increasing tenant protection may overcome an inefficient segmentation of good (bad) tenants in short-term (long-term) tenancies in market equilibrium. Hubert (1995) discusses the combination of rent control and tenant protection to prevent economic eviction. The analysis implies that the length of residency discount increases with the strength of rent control. However, the lower the rent the stronger is the incentive for economic eviction, possibly using one of the legal routes (e.g. modernization of apartment). Altogether it is an open empirical question as to whether tenancies with low rents are more likely to survive because of the higher interest of tenants to keep a cheap apartment or less likely to survive because of the higher interest of landlords in economic evictions. Furthermore, rent control may be binding more for the rent increases for new leases of more expensive apartments. Thus, it is an open empirical question as to how the length of residency discount varies across the distribution of rents.

Hoffmann and Kurz (2002) find a length of residency discount for Germany which could be a kind of compensation for the diminishing quality of an apartment over time. Schlicht (1983) interprets the discount as a landlord's concession trying to keep good tenants, especially when tenants' preferences change over time and landlords want to avoid turnover costs, e.g. forgone rents and search costs for new tenants. The existing empirical literature for the U.S. mostly finds evidence for a length of residency discount in average real rents for rental housing when regulation is lower than in Germany, see e.g. Barnett (1979), Börsch-Supan (1994), Follain and Malpezzi (1980), Noland (1979), Goodman and Kawai (1985), Basu and Emerson (2000), or Guasch and Marshall (1987). Guasch and Marshall (1987) decompose the discount into a *pure sit discount* and a *length of residency discount*. While the former discount is offered by landlords when contracts are renegotiated, the latter discount is given for each additional year tenants spend in the same rental unit. Using the Annual Housing Survey (AHS) data from 1974 to 1977, they estimate multiple specifications and find a sit discount between 6% and 13% and an annual residency discount between 0.2% and 0.8%.¹⁰ Using data of 102 apartment complexes in US-American metropolitan areas, Barker (2003) finds that discounts for short-term tenants are more common. Since rental payments rise faster than turnover costs, he predicts

¹⁰Guasch and Marshall (1987) also implement a selection correction for the termination of a tenancy accounting for selection on unobservables. Correcting for selection on unobservables in our quantile regression estimates is beyond the scope of this paper for two reasons. First, how to account for selection when estimating quantile regression is still subject of an intensive debate (Huber and Melly 2012). Second, finding a credible instrument with sufficient bite is difficult.

that discounts for new leases become more frequent. Sims (2007) analyzes the effect of rent control in various cities in Massachusetts that ended in 1995. The rent increase was adjusted to a specific annual rate, condominium conversions was made harder for landlords to avoid a reduction of the rental stock, and a prohibition to evict tenants without permission was imposed. Altogether, only 20% of rental housing was under active control because vacancy decontrol was possible. Based on difference-in-differences estimates, Sims (2007) finds that rent control leads to a significant rent decline. Furthermore, tenants' mobility falls as measured by significantly longer tenancies and the stock of rental housing declines because of the reduced attractiveness of rental apartments as investments for landlords.

Summing up and providing an outlook on our empirical analysis, our reading of the literature implies that the empirical studies so far have been restricted to an empirical analysis of how average rents vary by length of residency and other characteristics of the apartment and the tenant. We provide an analysis of the change in rents for new leases and of the effect of the length of residency (elapsed tenure) depending on the level of rents using quantile regression. The theoretical considerations above suggest that the length of residency discount depends upon tenants' characteristics such that tenants with lower mobility costs experience a higher discount. The discount should be larger when rents for new leases show a strong upward trend. It is an open empirical question as to how the length of residence discount varies with the level of rents because, on the one hand, tenants are more interested to keep a cheaper apartment and rent control may be binding more for the rent increases for new leases of more expensive apartments. On the other hand, landlords of cheaper apartments have a stronger incentive for economic evictions and there may be a stronger need for a modernization of the apartment justifying a rent increase. Furthermore, the above considerations imply that a reform strengthening the protection of sitting tenants against eviction and against rent increase should increase the length of residency discount. However, it is a priori unclear as to how the increase varies with the rent level. Because the theoretical considerations suggest that the rent level and the length of residency discount depend both upon tenants' characteristics and characteristics of the apartment, we use a panel of linked housing-tenant data.

3 Data

The empirical analysis uses the German Socio-Economic Panel Study (SOEP), a representative annual household panel survey.¹¹ Because of a lack of information on East German rental units before German unification and the ongoing transition process as well as the strong regulation of rents in East Germany during the 1990s, our analysis is restricted to West Germany.

The SOEP offers detailed information on rental housing from the perspective of tenants, thus providing *linked housing-tenant data*.¹² Because these are panel data on tenancies, we can study the length of residency discount within tenancies. The available variables include the size of an apartment (or house) in square meters (sqm), its equipment like the existence of a basement, balcony or terrace, and a garden, the type of building regarding the number of rental units and the year of construction, information as to whether the apartment is subsidized by the government and as to whether there is a private or a public landlord. We exclude outliers with a reported apartment size of less than 20 sqm and of more than 200 sqm as well as observations with a reported rent of less than 50 Euros (in current prices).

To account for variation in regional housing markets, we account for the state ("Bundesland") and we use detailed information on the location available in the SOEP, such as city size (number of inhabitants), region, type of residential area, and information on amenities in the local neighborhood.¹³ In addition, one observes the length of residence so far (elapsed tenure \equiv elapsed tenancy duration). Our panel data allow to control for unobserved time-invariant tenancy-specific characteristics, which can account for biases induced by the selective termination of tenancies (Guasch and Marshall 1987). Our dependent variable of interest is the real gross rent actually paid (without costs for heating) - henceforth referred to as rent. We deflate rents to 2005 prices using the consumer price index.

The final data set consists of 11,328 households and 18,601 tenancies, which means

¹¹We use the version of the data set for the time period 1984-2011, http://dx.doi.org/10. 5684/soep.v28.1 (see Wagner et al. (2007), Wagner et al. (2008) and Schupp (2009) for further information).

 $^{^{12}}$ We coin this term in analogy to linked employer-employee data used in labor market research. Up to our knowledge, the term *linked housing-tenant data* (or another term conveying the same idea) has not been used so far in the literature.

¹³The information on the amenities is available for the years 1986, 1994, 1999, 2004, and 2009. To impute the values in between, we assume that within a tenancy the distance to amenities measured by the time needed by foot to reach the amenity does not change over time.

that we observe on average 1.6 tenancies per household. To account for the 2001 Tenancy Law Reform Act, we define a dummy variable that indicates whether the tenancy started on September 1, 2001 or later. To make the samples before and after the reform comparable, we restrict the panel data set to those observations with elapsed tenure up to 10 years because we can not observe a longer tenure after the reform.

4 Econometric Approach

We estimate a standard hedonic price regression which we augment with variables observed for tenants.¹⁴ The estimated coefficients of the apartment characteristics are interpreted as implicit prices, which may be interpreted as the consumer's marginal willingness to pay, if he or she is able to choose between a sufficiently large number of units that vary in their characteristics.

Based on the limited set of characteristics of the apartment observed and because of likely frictions in the rental housing market, prices for apartments with the same observed characteristics do vary.¹⁵ Consumers may differ in their willingness to pay and the limited mobility of tenants may prevent relative prices to equal the willingness to pay for certain characteristics. These issues motivate the estimation of quantile regressions. In addition, it is likely that the distribution of unobservables, as measured by the dispersion of prices within cells defined by observed apartment characteristics, may differ across cells.

4.1 Specification and Identification

We specify the rent for rental unit i in year t by

(1)
$$log(rent_{it}) = \sum_{j=1}^{k} \beta_j \cdot X_{j,i,t} + f(Ten_{it})'\gamma + \delta_0 RD_{it} + [RD_{it}f(Ten_{it})]'\delta_1 + \mu_t + \varepsilon_{it}$$

¹⁴Waugh (1929) and Court (1939) were the first to use this approach. Griliches (1961 and 1971), Lancaster (1966 and 1971), and Rosen (1974) introduced the hedonic price model to a wider audience of economists.

¹⁵Sirmans, Macpherson, and Zietz (2005), Zietz, Zietz, and Sirmans (2007) and Zietz, Sirmans, and Smersh (2008) address the heterogeneity of implicit prices in housing markets. Decomposing the unconditional rental price distribution of advertised apartments in Berlin, Thomschke (2015) finds that the increase in rental prices are due to a changing demand structure regarding quality and quantity rather than a change in apartment characteristics, especially in the high-price segment.

where $log(rent_{it})$ denotes the log deflated rental payment, $X_{j,i,t}$ for $j = 1, \ldots, k$ are the apartment characteristics and β_j the corresponding implicit price. Ten_{it} is the elapsed tenure of unit *i* in year *t* and our specification will use linear splines in elapsed tenure. $f(Ten_{it})$ involves the specific terms of the separate spline segments and γ and δ_1 are the corresponding coefficient vectors, respectively. μ_t are year fixed effects. RD_{it} is a reform dummy which is zero for tenancies starting before the 2001 reform and one for tenancies starting after the 2001 reform. ε_{it} is the idiosyncratic error term. For our fixed effects estimates, tenancy specific fixed effects are added to equation (1).

The identification of the reform effects in equation (1) relies on the differences between tenancies starting before 2001 and still continuing after 2001 and tenancies starting after 2001. The reform only applies for those tenancies starting after 2001. Thus, the year effects starting from 2001 onwards can be separated from the reform dummy because the reform effect does not apply for tenancies starting before 2001 and continuing at least until 2001. Analogously, we can identify the coefficients of the reform dummy and its interactions with the linear tenancy splines. For the OLS fixed effects estimator, the reform dummy cannot be identified. We also do not include a reform dummy in the panel fixed effects quantile regressions as described in the following. However, the coefficients of the interactions of the linear tenancy splines with the reform dummy are identified because, over time, tenancies do change from one spline segment to the next.

To identify the nonlinear effect of elapsed tenure on rental payment, we construct linear splines. After a preliminary cross-validation of models with equally positioned but varying number of knots, the preferred specification has one knot at three years of elapsed tenure.

4.2 Quantile Regression

Quantile regressions allow to estimate how the market valuation of characteristics of apartments varies with the level of rents across the conditional rent distribution (see Koenker and Hallock 2001 and Koenker 2005 for details). A further advantage is that quantile regressions are more robust than OLS to outliers in the dependent variable. Estimating quantile regressions for panel data within a tenancy can reveal the net effect of rent setting for new leases and sorting effects due to termination of tenancies if the ranking of rents across tenancies does not change over time. For instance, if the length of residency discount is generally higher (lower) at the top of the rent distribution than at the bottom, then tenancies for cheaper apartments may be more (less) likely to end early. Furthermore, if the discount between the first and the second year of a tenancy is higher (lower) at the top of the rent distribution than at the bottom, then rents for new leases grow more (less) for more expensive apartments compared to less expensive apartments.

Using the same specification as for OLS, we estimate quantile regressions at the median, at the 17%- (one sixth) and at the 83%- (five sixth) quantile of the conditional rent distribution. This way we cover the two-third span of conditional rents as it is customary for an official local rent index in Germany. For our baseline model, we also provide quantile regression estimates at each decile to investigate whether the results at the three quantiles we focus upon are representative for the entire distribution. We obtain clustered standard errors for our estimated coefficients through bootstrapping based on 200 resamples. We resample entire tenancies to account for heteroscedasticity and serial correlation of the error term within a tenancy.

As robustness check, we estimate panel regressions accounting for mean tenancy fixed effects as suggested for quantile regression by Canay (2011).¹⁶ We first estimate the fixed effects OLS regression and obtain the mean tenancy fixed effects. Then, to implement the quantile regressions with fixed effects, we substract the mean tenancy fixed effects from the rents within a tenancy and estimate the panel quantile regressions for these adjusted rents based on an intercept and the time-varying covariates. Note that the approach does not allow the fixed effects to differ across quantiles. Accounting for quantile specific fixed effect would result in an estimator which is not easy to interpret because then the conditional distribution for a given tenancy modeled by the quantile regression estimates would change by quantile. In contrast to fixed effects OLS estimation, our fixed effects quantile regression still suffers from the incidental parameter problem even though the estimated fixed effect is part of the dependent variable.¹⁷ It is likely that our fixed effects quantile

¹⁶We prefer this simple estimator compared to more involved alternatives proposed in the literature. Koenker (2004) was the first to suggest a fixed effects quantile regression by adding a dummy for each cross-sectional unit and by including an L_1 -penalization to shrink the individual fixed effects to zero (see also Koenker 2005, chapter 8.7, and Lamarche 2010). This estimator involves the choice of a shrinkage (tuning) parameter. Consistency of this estimator requires that both the number of cross-sectional units and the number of time periods converge to infinity. As an alternative approach, among others, Galvao and Kato (2015) suggest to estimate the asymptotic bias of the fixed effects quantile regression without penalization in order to correct for this bias.

¹⁷Note that the number of panel observations in our application is unlikely to be large enough to render the bias in the fixed effects estimator negligible. The average number of panel observations

regression estimator is smoothed across quantiles, because the estimation error in the fixed effects may attenuate the differences in coefficients across quantiles.¹⁸

5 Empirical Results

5.1 Descriptive Statistics

Table 1 shows means and standard deviations of apartment characteristics for the full sample as well as for subsamples with tenancies starting before September 1, 2001, or afterwards [before and after the reform - here and henceforth, *after September 1*, 2001 also includes tenancies that start on September 1, 2001]. Recall that to make the samples before and after the reform comparable, the samples are restricted to those observations with the elapsed tenure up to 10 years. 25% (75%) of the tenancies are observed to have started after (before) September 1, 2001. The column labeled "Difference" displays results of *t*-tests of equality of means for each variable in both subsamples. The average rent for tenancies starting after the reform is \in 461 and the average rent for tenancies starting before the reform is \notin 420.

Both subsamples differ significantly in further apartment characteristics. Tenancies starting after the reform are on average 2.4 sqm larger, can rather be found in more recently built tenancy-occupied houses (especially in houses built after 1990), and are on average better equipped (more likely to include a balcony, a terrace or a garden, and central heating). Mechanically, the elapsed tenure in tenancies starting after the reform is 1.5 years shorter. Also, the households in the two subsamples differ slightly in their assessment (assess) of the rental payment and the apartment size.^{19'20} Tenants in the before-reform subsample are 6.5 percentage points (ppoints) more likely to assess their rent to be inexpensive and they are 1.8 ppoints more likely

per tenancy in our sample (elapsed tenure up to 10 years) is 5.6. The lower quartile is 3 and the upper quartile is 8. Still, if the fixed effects quantile regression estimates differ from the standard panel quantile regression estimates, the direction of the change is likely to be informative, especially if it corresponds to the OLS fixed effects estimator.

¹⁸We are extremely grateful to a referee to point this out and to refer us to an unpublished study. Unfortunately, this study is very preliminary and at this point we are not supposed to provide the reference.

¹⁹The variables displayed in Table 1 under "Further Characteristics" are only used for descriptive comparisons. They are not part of the specification of the final regressions.

 $^{^{20}}$ The differences in assessment we observe are larger when adding observations with tenure above 10 years to the before-reform sample. The evidence is consistent with a tenancy being more likely to last if the renter is more satisfied, i.e. better matches are more likely to last. These results are available upon request. We are grateful to a referee for suggesting this point.

to assess their apartment to be too small. At the same time, the rent-to-income ratio increases by 2.7 ppoints for the after-reform sample.²¹ Overall, the descriptive evidence suggests that the demand for larger (and better equipped) apartments is rising over time and households are willing to spend a higher share of their income on rents. However, the difference could also reflect the remaining difference in elapsed tenure in combination with better matches being more likely to last.

— Insert Table 1 here. —

Table 2 shows mean differences in some key variables between the first tertile (lowprice segment) and the third tertile (high-price segment) of the unconditional rent distribution (recall that rents are deflated by the CPI to 2005 prices). Apartments in the high-price segment are 34 sqm larger than in the low-price segment. High-price apartments are rather located in recently built houses/buildings, in new residential areas, in larger cities, and in city centers (evidence on these variables is available upon request). As to be expected, high-price apartments are on average better equipped. Furthermore, apartments in the high-price segment are rather located next to stores, parks, sports complexes or public transport, and the apartment size per person is larger. Households in the high-price segment tend to assess the rent as being too high and the size of the apartment as being too large compared to the assessments in the low-price segment. Consistent with the subjective assessments, there is a 7 ppoints higher rent-to-income ratio in the high-price segment. The evidence suggests that tenants in the high-price segment demand higher quality apartments, but they tend to think that their apartments are over-priced and possibly too well equipped relative to their needs. These findings may rationalize the political momentum towards stronger rent control in the wake of the large rent increases during the 1990s.

— Insert Table 2 here. —

5.2 Estimation Results

Table 3 provides the estimation results for our baseline panel OLS and quantile regressions. We estimate quantile regressions at the median (QR 50%), at the one

 $^{^{21}}$ The rent-to-income ratio is calculated as the share of the household net income that is spent on the gross rent without heating.

sixth (QR 17%), and at the five sixth (QR 83%) quantile. The column '83%-17%' involves t-statistics for the significance of the difference of the coefficients between QR 83% and QR 17%.²²

Quite uniformly, the covariates apartment size, city size, and central location show the expected positive effects on rents. The differences between the two tail quantiles are mostly not significant, but there are some notable exceptions. The average partial effect (APE) of apartment size increases between the two quantiles by 4.1 log points, thus implying a higher dispersion of rents for larger apartments. Regarding the type of house, rents for nondetached houses and apartments in multiunit buildings are ceteris paribus higher than in detached houses. A priori, one would expect a higher rent for detached houses and our estimates should not be viewed as being causal. Most likely, detached houses and multiunit buildings differ on average in some unmeasured characteristics. Furthermore, in multiunit buildings, the rents are less dispersed than in detached houses (the quantile differences are significantly negative). It is likely that unobserved characteristics among detached houses are more dispersed than in more standardized multiunit buildings. Rents are higher in new residential areas compared to old residential areas, typically in apartments with more amenities in walking distance, and in better equipped apartments (the results for the latter three variables are available upon request). Rents are higher and more dispersed for private landlords and for non-subsidized housing. A housing subsidy may imply a particular restriction on rent increases for higher rents. Private landlords, who tend to focus more on higher revenues, are more likely to raise rents compared to non-private landlords.

- Insert Table 3 here. -

Let us now turn to our estimates of the length of residency discount. We estimate flexible linear splines and our regressions report the slope (the annual discount) for each linear segment of our splines. For the first three years of a tenancy, we find a significant annual reduction of rents which amounts to 1.2 log points per year and the decline is significantly stronger for higher rents. At the 83% quantile the

²²We also estimate unconditional quantile (RIF) regressions (Firpo et al. 2009) to investigate the effects on the unconditional rent distribution. The reform effect and the length of residency discounts are quite similar to those estimated by the conditional quantile regressions reported in this paper. However, the estimated coefficients are in some instances more heterogeneous across quantiles. The results for the RIF regressions are available upon request. Because the level of rents changes strongly over time, our study focuses on conditional quantile regression estimates.

annual discount amounts to 1.7 log points while at the 17% quantile the annual discount is only 0.7 log points. From the fourth year of a tenancy onwards, the residency discounts become more uniform. While at the 17% quantile the annual discount increases slightly to 0.8 log point, it falls to 0.9 log points at the 83% quantile. Figure 3 provides the estimated annual discounts at each decile confirming that the results at the three quantiles reported in the table are representative for the entire distribution. We find that the dispersion of rents falls during the first years of a tenancy while there is no evidence for a significant further reduction of the dispersion after the fourth year.

The baseline rent regressions control for year dummies, states, and the year of construction (the corresponding coefficients and further details are available upon request). As to be expected, rents increase with the year of construction and the increase is significantly stronger at the bottom of the rent distribution compared to the top. There are also significant differences in rents across states.

Next we turn to the time effects and the reform effects. The estimated time effects reflect a uniform growth of average and median rents (deflated by the CPI) between 1984 and 2001 by 26 log points. The increase is significantly higher by about seven log points at the 17% quantile compared to the 83% quantile. The stronger increase at the bottom of the distribution may rationalize the strong political demands for further rent control in the 1990s and 2000s. Incidentally, rents do not increase further after 2001 and rents at the 17% quantile (83% quantile) are six (two) log points lower in 2011 compared to 2001. Apparently, rent growth has stopped after 2001 but it has to be kept in mind that the regression also includes a dummy variable 'After Reform' which corresponds to tenancies starting after the reform in 2001. The OLS reform effect is minus four log points and the decline is about three log points stronger at the top of the distribution compared to the bottom. Figure 4 shows that the estimates by decile confirm a uniformly growing reform effect (in absolute value) along the distribution. These findings suggest a stronger effect of the reform in curtailing higher rents. However, the difference between the two tail quantiles is not significant. Altogether we find a general decline in rents after 2001 and a further decline of rents for new tenancies after the reform. While the general decline is stronger for lower rents, the specific (partial) reform effect is stronger for higher rents. Prima facie, our findings suggest that in times of generally falling rents the reform did in fact result both in reducing rent growth for new leases, especially for expensive new leases. The latter implies that households living in

expensive apartments tend to benefit more from the reform than households living in cheaper apartments. Thus, one may be concerned that the reform may not have been sufficiently targeted, if the goal was to curtail rents for low-income tenants who tend to pay lower rents.

Now, we investigate the impact of the reform on the residency discount. One could expect that the reform increases the discount, and the reduction should be particularly strong for higher rents. We keep the reform dummy and we add interactions of the reform dummy with the splines in elapsed tenure. Table 4 provides the annual discounts estimated separately before and after the reform (all other covariates are as in the baseline regressions reported in Table 3 and results for the other covariates are available upon request). The level effect of the after reform dummy now shows a particularly strong negative reform effect on rents for new leases of about ten log points on average. The effect is significantly larger in absolute value at the top compared to the bottom of the distribution. Before the reform, during the first three years of a tenancy, the annual discount is 2.5 log points at the 83% quantile, being 1.2 log points (significantly) larger than at the 17% quantile. From the fourth year onwards, the annual discount is more uniform and lies between 0.8 and 0.9 log points. After the reform, the annual discount becomes insignificant and the point estimates are mostly positive. The difference (after minus before) is significantly positive and the reduction in the discount is higher at the bottom compared to the top of the distribution. Basically, after the reform there is no residency discount any more. In addition to the changes in the coefficients, we also calculate the APE of the reform effect during the two tenure intervals which table 4 reports as 'After Reform (APE)'.²³ We find a significantly negative average reform effect during the first three years of a tenancy which increases along the distribution. However, there is no significant average reform effect after the third year of a tenancy. Our overall findings may be related to the fact that in general real rents are falling after the reform and CPI inflation is generally low. Somewhat in contrast to our prior expectation, the rent gap between new leases and sitting tenants does not increase after the reform. The reform effect on new leases is particularly strong and rent increases fall in general after the reform, which eliminates the length of residency discount because nominal rents are unlikely to fall. These are surprising findings because the reform attempted to curtail rent increase for sitting tenants and it did not change regulations of the rents for new leases.

 $^{^{23}}$ The APEs add the level effect of the reform dummy and the effect on the cumulative discount.

- Insert table 4 here. -

5.3 Tenancy Fixed Effects

To examine the sensitivity of our results to potential unobserved confounders that are constant within tenancies, we reestimate the model in Table 4 also accounting for tenancy fixed effects. To do so, we first exclude 5,870 tenancies with only one observation and estimate the tenancy fixed effects regression based on this restricted sample. Table 5 provides the fixed effects estimates for the residency discount before and after the reform. Note that we can not estimate the level effect of the reform, when accounting for tenancy fixed effects when accounting for tenancy fixed effects.

The estimated annual discounts differ from the results reported in Table 4. They are reduced before the reform and they even become significantly positive after the reform. Before the reform, the annual discount during the first three years of tenure is only 0.4 log points for OLS, 0.2 log points at the 17% quantile, and 0.6 log points at the 83% quantile. However, before the reform for elapsed tenure above three years and after the reform the discount turns significantly positive for OLS for all levels of elapsed tenure as well as for the median and the upper part of the distribution after three years of elapsed tenure. These changes between the estimates in Tables 4 and 5 are compatible with sorting effects such that high price tenancies are more likely to end early, thus enhancing the 'observed' discount as reported in Table 4. The finding is consistent with tenants in high-price rental units searching more strongly for cheaper alternatives, partly because rents in these tenancies are being increased over time. In contrast, tenancies with a low rent tend to last longer. These findings are not consistent with the hypothesis that tenants in the low-price segment are more likely to experience economic evictions or to search for better quality alternatives. After the reform, landlords seem to increase rents within a tenancy in a low-inflation environment, which is still consistent with the allowed maximum rent increases for sitting tenants and which may be a compensation for the drop in rents for new leases. Landlords may be using their market power arising from the cost of moving to a new apartment. Considering the differences in the residency discount induced by the reform, we find a similar direction of the effects in Tables 4 and 5, but the size of the estimated changes is reduced when accounting for fixed effects.

- Insert table 5 here. -

Summing up, controlling for tenancy fixed effects changes the estimates for the residency discounts, effectively revealing that the discount is smaller before the reform, and we even find significantly positive tenure effects on rents after the reform. The evidence is consistent with expensive tenancies being likely to end earlier and new leases benefitting strongly from a reduction in rents induced by the reform. It is a robust finding that the reform reduces the residency discount and that there is no evidence for a residency discount after the reform.

6 Conclusions

The large private rental housing market in Germany shows a higher level of regulation through rent control and tenant protection against eviction than the OECD average. In contrast to the current situation, Germany was characterized by fairly liberal rent laws until the 1990s. In the wake of housing shortages and strong rent increases in the 1990s, there was a change towards more regulation in order to protect sitting tenants. For instance, the German government passed the Tenancy Law Reform Act in 2001 to restrict rent increases and to strengthen the protection of tenants against eviction. Based on linked housing-tenant data from the Socioeconomic Panel, this paper estimates panel OLS and quantile regressions of rents within tenancies during the time period 1984 to 2011. Specifically, we analyze the impact of the Tenancy Law Reform Act implemented in 2001 (the reform) on the level of rents and on the residency discount.

Our findings can be summarized as follows. Rents deflated by the CPI increase strongly from 1984 until the reform in 2001, and there is a reversal in the trend afterwards. Households whose tenancy started after the reform rather live in newer, better equipped, and larger rental units. Both the level of rents and the rent-toincome ratio grow after the reform. This suggests that demand for better equipped housing increases over time.

Before the reform, we find a significant residence discount in the rent level which decreases in absolute value with tenure and which is stronger at the top than at the bottom of the rent distribution. The reform reduces rents, in particular for expensive apartments and for new leases. The reform eliminates the residency discount and the fixed effects estimates reveal that after the reform rents in fact grow with tenure, especially for more expensive apartments. These are surprising findings because the reform attempted to curtail rent increase for sitting tenants and it did not change regulations of the rents for new leases. The evidence is consistent with expensive tenancies being likely to end earlier and tenants in new leases benefitting strongly from a reduction in rents induced by the reform. After the reform, in a low-inflation environment, landlords seem to increase rents within a tenancy which is still consistent with the allowed maximum rent increases for sitting tenants and which may be a compensation for the drop in rents for new leases. Landlords may be using their market power arising from the cost of moving to a new apartment. Altogether, households living in expensive apartments tend to benefit more from the reform than households living in cheaper apartments.

Our evidence suggests that the reform was successful in curtailing rent increases for new leases and for expensive apartments early in a tenancy. It remains an open question as to whether this led to fewer houses being built, which may imply detrimental long-run effects for renters.

References

- Barker, David (2003). "Length of Residence Discounts, Turnover, and Demand Elasticity. Should Long-Term Tenants Pay Less than New Tenants?", *Journal* of Housing Economics, Vol. 12, 1-11.
- Barnett, Lance (1979). "Using Hedonic Indexes to Measure Housing Quality", *The RAND Corporation*, Santa Monica, CA, R-2450-HUD.
- Basu, Kaushik and Patrick M. Emerson (2000). "The Economics of Tenancy Rent Control", *The Economic Journal*, Vol. 110(466), 939-962.
- Bischoff, Oliver and Wolfgang Maennig (2011). "Rental housing market segmentation in Germany according to ownership", *Journal of Property Research*, Vol. 28(2), 133-149.
- Börsch-Supan, Axel (1986). "On the West German Tenants' Protection Legislation", Journal of Institutional and Theoretical Economics, Vol. 142(2), 380-404.
- Börsch-Supan, Axel (1994). "Housing Market Regulations and Housing Market Performance in the United States, Germany, and Japan", in: Blank, Rebecca M. "Social Protection versus Economic Flexibility: Is There a Trade-Off?", National Bureau of Economic Research, University of Chicago Press, pages 119-156.

- Bundestag (2000). "Entwurf eines Gesetzes zur Neugliederung, Vereinfachung und Reform des Mietrechts (Mietrechtsreform), Drucksache 14/4553" (accessed on March 11, 2014), Retrieved from http://dip21.bundestag.de/dip21/btd/ 14/045/1404553.pdf
- Canay, Ivan A. (2004). "A simple approach to quantile regression for panel data", *The Econometrics Journal*, Vol. 14(3), 368-386.
- CESifo (2005). "Tax Preferences for Housing", *CESifo DICE Report 2005*, vol. 3(1), 83-84.
- Court, Andrew T. (1939). "Hedonic Price Indexes with Automotive Examples", *The Dynamics of Automobile Demand*, General Motors, 99-119.
- Court, Louis M. (1941). "Entrepreneurial and Consumer Demand Theories for Commodity Spectra", *Econometrica*, Vol. 9(2), 135-162.
- Eeckhoff, Johann (1981). "Zur Kontroverse um die ökonomische Auswirkungen des Zweiten Wohnraumkündigungsschutzgesetzes", Journal of Institutional and Theoretical Economics, Vol. 137(1), 62-77.
- Eurostat (2013). Distribution of population by tenure status in 2013. Retrieved from http://epp.eurostat.ec.europa.eu/statistics_explained/index. php/Housing_statistics [acessed on October 22, 2014, online data code: ilc_lvho02].
- Firpo, Sergio, Nicole Fortin, and Thomas Lemieux (2009). "Unconditional quantile regressions", *Econometrica*, Vol. 77(3), 953-973.
- Follain, James R. and Stephen Malpezzi (1980). "Dissecting Housing Value and Rent: Estimates of Hedonic Indexes for Thirty-Nine Large SMSAs", *The Urban Institute*, Washington D.C.
- Galvao, Antonio and Kengo Kato (2015). "Smoothed Quantile Regression for Panel Data", Unpublished Manuscript, University of Iowa, Available at SSRN 2606188.
- Goodman, Allen C. and Masahiro Kawai (1985). "Length-of-Residence Discounts and Rental Housing Demand: Theory and Evidence", *Land Economics*, Vol. 62(2), 93-105.
- Griliches, Zvi (1961). "Hedonic Price Index for Automobiles: An Econometric Analysis of Quality Change", The Price Statistics of the Federal Government, No.73.
- Griliches, Zvi (1971). "Price Indexes and Quality Change: Studies in New Methods of Measurement", *Havard University Press*, Cambridge.

- Guasch, José L., and Robert C. Marshall (1987). "A Theoretical And Empirical Analysis of the Length of Residency Discount in the Rental Housing Market", *Journal of Urban Economics*, Vol. 22, 291-311.
- Hoffmann, Johannes, and Claudia Kurz (2002). "Rent Indices for Housing in West Germany 1985 to 1998", Deutsche Bundesbank Economic Research Centre Discussion paper 01/02, Frankfurt a.M.
- Huber, Martin and Blaise Melly (2012). A test of the conditional independence assumption in sample selection models. Unpublished Manuscript, Brown University.
- Hubert, Franz (1995). "Contracting with costly tenants", *Regional Science and Urban Economics*, vol. 25(5), 631-654.
- Hubert, Franz (1998). "Private Rented Housing in Germany", Neth. J. of Housing and the Built Environment, Vol. 13(3), 205-232.
- Johansson, Asa (2011). "Housing Policies in OECD Countries: Survey-Based Data and Implications", OECD Economics Department Working Papers No. 836.
- Koenker, Roger and Kevin F. Hallock (2001). "Quantile Regression", Journal of Economic Perspectives, Vol. 15, 143-156.
- Koenker, Roger (2005). "Quantile Regression", *Cambridge University Press*, New York.
- Koenker, Roger (2004. "Quantile Regression for Longitudinal Data", Journal of Multivariate Analysis, Vol. 91(1), 74-89.
- Lamarche, Carlos (2010). "Robust penalized quantile regression estimation for panel data". Journal of Econometrics, Vol. 157(2), 396-408.
- Lancaster, Kevin J. (1966). "A New Approach to Consumer Theory", Journal of Political Economy, Vol. 74(2), 132-157.
- Lancaster, Kevin J. (1971). "Consumer Demand: A New Approach", *Columbia University Press*, New York.
- Noland, Charles W. (1979). "Assessing Hedonic Indexes for Housing", The Journal of Financial and Quantitative Analysis, Vol. 14(4), 783-800.
- Rosen, Sherwin (1974). "Hedonic Prices and Implicit Markets: Product Differentiation in Pure Competition", *Journal of Political Economy*, Vol. 82(1), 34-55.
- Schlicht, Ekkehart (1983). "The Tenant's Decreasing Willingness to Pay and the Rent Abatement Phenomenon", Journal of Institutional and Theoretical Economics, Vol. 139, 155-159.

- Schwager, Robert (1994). "On the West German Tenants' Protection Legislation: A Comment", Journal of Institutional and Theoretical Economics, Vol. 150(2), 411-418.
- Schupp, Jürgen (2009). "25 Jahre Sozio-oekonomisches Panel Ein Infrastrukturprojekt der empirischen Sozial- und Wirtschaftsforschung in Deutschland", Zeitschrift für Soziologie, 38 (5), 350-357.
- Sheppard, Stephen (1999). "Hedonic analysis of housing markets", in: Paul C. Cheshire and Edwin S. Mills (ed.), Handbook of Regional and Urban Economics, Vol. 3, ch. 41, pages 1595-1635.
- Sims, David P. (2007). "Out of Control: What Can We Learn from the End of Massachusetts Rent Control?", Journal of Urban Economics, Vol. 61(1), 129-151.
- Sirmans, George S. David A. Macpherson, and Emily N. Zietz (2005). "The Composition of Hedonic Pricing Models", *Journal of Real Estate Literature*, Vol 13(1), 1-44.
- Thomschke, Lorenz (2015). "Changes in the Distribution of Rental Prices in Berlin", *Regional Science and Urban Economics*, Vol. 51, 88-100.
- Wagner, Gert G., Joachim R. Frick, and Jürgen Schupp (2007). "The German Socio-Economic Panel Study (SOEP): Scope, Evolution and Enhancements", *Schmollers Jahrbuch*, 127 (1), 139-169.
- Wagner, Gert G., Jan Göbel, Peter Krause, Rainer Pischner, and Ingo Sieber (2008). "Das Sozio-oekonomische Panel (SOEP): Multidisziplinäres Haushaltspanel und Kohortenstudie für Deutschland – Eine Einführung (für neue Datennutzer) mit einem Ausblick (für erfahrene Anwender)", AStA Wirtschafts- und Sozialstatistisches Archiv, 2 (4), 301-328.
- Waugh, Frederick van (1929). "Quality as a Determinant of Vegetable Prices", *Columbia University Press*, New York.
- Zietz, Joachim, Emily N. Zietz, and George S. Sirmans (2007). "Determinants of House Prices: A Quantile Regression Approach", Journal of Real Estate Finance and Economics, Vol. 37(4), 317-333.
- Zietz, Joachim, George S. Sirmans, Greg T. Smersh (2008). "The Impact of Inflation on Home Prices and the Valuation of Housing Characteristics Across the Price Distribution", Journal of Housing Research, Vol. 17(2), 119-137.

Figures



Figure 1: Rent control in the private rental market,¹ 2009 Scale 0-6: Increasing in degree of control

Figure 2: Tenant-landlord regulations in the private rental market,² 2009 Scale 0-6: Increasing in protection for tenants



- 1. This indicator is a composite indicator of the extent of controls of rents, how increases in rents are determined and the permitted cost pass-through onto rents in each country. Control of rent levels includes information on whether rent levels can be freely negotiated between the landlord and the tenant, coverage of controls on rent levels and the criteria for setting rent levels (market based, utility/cost based, negotiation based or income based). Controls of rent increases includes information on whether rent increases can be freely agreed by the landlord/tenant, whether rent increases are regularly indexed to some cost/price index or if increases are capped or determined through some other administrative procedure, including negotiation between tenant/landlord associations. The pass-through of costs onto rents includes information on whether reallowed to pass on increases in costs onto rents (cost pass-through) and the extent of such pass-through i.e. the types of cost that can be passed on.
- 2. The indicator measures the extent of tenant-landlord regulation within a tenancy. It includes the ease of evicting a tenant, degree of tenure security and deposit requirements.

Source: Johansson (2011)





Note: Quantile regression estimates for baseline model 1 as specified in Table 3.

Figure 4: After Reform - Coefficient Estimate by Quantile



Note: Quantile regression estimates for baseline model 1 as specified in Table 3.

Tables

| Rent (Deflated to 2005 €) 429.786 419.522 460.864 41.342** (208.231) (201.072) (225.720) (1.902) Apartment Size (in m ²) 72.530 71.933 74.338 2.404** Elapsed Tenure (in years) 3.802 4.165 2.705 -1.460** (2.790) (2.903) (2.060) (0.025) Private Landlord 0.789 0.789 0.000 (0.408) (0.408) (0.408) (0.408) (0.004) Subsidized Housing 0.110 0.131 0.047 -0.084** (0.313) (0.338) (0.211) (0.003) 1918 to 1948 0.165 0.168 0.154 -0.014** (0.371) (0.374) (0.361) (0.004) 1949 to 1971 0.372 0.380 0.345 -0.034** (0.483) (0.485) (0.475) (0.005) 1972 to 1980 0.198 0.209 0.164 -0.044** (0.253) (0.248) (0.268) (0.022) Since 1991 0.0399 (0.406) (0.370) (0.003) </th <th>Variable</th> <th>Full</th> <th>Before Reform</th> <th>After Reform</th> <th>Difference</th> | Variable | Full | Before Reform | After Reform | Difference |
|--|--|-------------|---------------------------|----------------|----------------|
| $\begin{array}{ccccccc} (208.231) & (201.072) & (225.720) & (1.902) \\ \text{Apartment Size (in m2)} & 72.530 & 71.933 & 74.338 & 2.404** \\ & (26.613) & (26.220) & (27.690) & (0.2404* \\ & Elapsed Tenure (in years) & 3.802 & 4.165 & 2.705 & -1.460** \\ & (2.790) & (2.903) & (2.060) & (0.025) \\ \text{Private Landlord} & 0.789 & 0.789 & 0.789 & 0.000 \\ & (0.408) & (0.408) & (0.408) & (0.004) \\ & \text{Subsidized Housing} & 0.110 & 0.131 & 0.047 & -0.084** \\ & (0.313) & (0.338) & (0.211) & (0.003) \\ \hline & & & & & & & & & & & & & & & & & &$ | Rent (Deflated to $2005 \in$) | 429.786 | 419.522 | 460.864 | 41.342*** |
| Apartment Size (in m ²) 72.530 71.933 74.338 2.404** (26.613) (22.220) (27.690) (0.244) Elapsed Tenure (in years) 3.802 4.165 2.705 -1.460** (2.790) (2.903) (2.060) (0.025) Private Landlord 0.789 0.789 0.789 0.000 (0.408) (0.408) (0.408) (0.004) Subsidized Housing 0.110 0.131 0.047 -0.084** (0.313) (0.338) (0.211) (0.003) Year of Construction 106 0.096 -0.011** (0.305) (0.308) (0.294) (0.003) 1918 to 1948 0.165 0.168 0.154 -0.014** (0.431) (0.455) (0.475) (0.005) 1972 to 1980 0.198 0.209 0.164 -0.044** (0.253) (0.248) (0.268) (0.002) Since 1991 0.093 0.071 0.163 0.092** (0.253) (0.248) (0.268) (0.002) Since 1991 0.0372 | | (208.231) | (201.072) | (225.720) | (1.902) |
| $\begin{array}{c ccccccccccccccccccccccccccccccccccc$ | Apartment Size (in m^2) | 72.530 | 71.933 | 74.338 | 2.404*** |
| Elapsed Tenure (in years) 3.802 4.165 2.705 -1.460^{**} (2.790) (2.903) (2.060) (0.025) Private Landlord 0.789 0.789 0.708 0.000 (0.408) (0.408) (0.408) (0.004) Subsidized Housing 0.110 0.131 0.047 -0.084^{**} (0.313) (0.338) (0.211) (0.003) Year of Construction Before 1918 0.104 0.106 0.096 -0.011^{**} (0.305) (0.308) (0.294) (0.003) 1918 to 1948 0.165 0.168 0.154 -0.014^{**} (0.371) (0.374) (0.361) (0.004) 1949 to 1971 0.372 0.380 0.345 -0.035^{**} (0.483) (0.485) (0.475) (0.005) 1972 to 1980 0.198 0.209 0.164 -0.044^{**} (0.253) (0.248) (0.268) (0.002) Since 1991 0.069 0.066 0.078 0.012^{**} (0.253) (0.248) (0.268) (0.002) Since 1991 0.0372 0.382 0.342 -0.041^{**} (0.253) (0.248) (0.268) (0.002) Since 1991 0.033 0.071 0.163 0.092^{**} (0.290) (0.257) (0.370) (0.004) 2 500k (Center) 0.372 0.382 0.342 -0.041^{**} (0.303) (0.314) (0.266) (0.003) 100k to 500k (Center) 0.153 0.134 0.211 0.077^{**} (0.360) (0.341) (0.408) (0.003) 100k to 500k (Suburb) 0.102 0.111 0.076 -0.034^{**} (0.360) (0.341) (0.408) (0.003) 100k to 500k (Suburb) 0.070 0.056 0.113 0.057^{**} (0.255) (0.229) (0.317) (0.002) 50k to 100k (Center) 0.033 0.032 0.025 -0.007^{**} (0.255) (0.229) (0.317) (0.002) 50k to 100k (Suburb) 0.026 0.019 0.048 0.288^{**} (0.161) (0.138) (0.213) (0.001) 20k to 500k (Suburb) 0.026 0.019 0.048 0.288^{**} (0.276) (0.273) (0.284) (0.033) 50k to 100k (Suburb) 0.026 0.019 0.048 0.288^{**} (0.276) (0.273) (0.284) (0.003) 50k to 100k (Suburb) 0.026 0.019 0.048 0.288^{**} (0.276) (0.273) (0.284) (0.003) 50k to 100k (Suburb) 0.026 0.019 0.048 0.288^{**} (0.276) (0.273) (0.284) (0.003) 50k to 100k (Suburb) 0.026 0.019 0.048 0.288^{**} (0.276) (0.273) (0.284) (0.003) 50k to 200k (Suburb) 0.026 0.019 0.048 0.288^{**} (0.276) (0.273) (0.284) (0.003) 50k to 200k (Suburb) 0.026 0.019 0.048 0.028^{**} (0.276) (0.273) (0.284) (0.003) 50k to 200k (Suburb) 0.026 0.019 0.048 0.028^{**} | | (26.613) | (26.220) | (27.690) | (0.244) |
| $\begin{array}{cccccccccccccccccccccccccccccccccccc$ | Elapsed Tenure (in years) | 3.802 | 4.165 | 2.705 | -1.460^{***} |
| Private Landlord 0.789 0.789 0.789 0.000 Subsidized Housing 0.110 0.131 0.047 -0.084** (0.313) (0.338) (0.211) (0.003) Year of Construction Before 1918 0.104 0.106 0.096 -0.011** (0.305) (0.308) (0.294) (0.003) 1918 to 1948 0.165 0.168 0.154 -0.014** (0.371) (0.371) (0.361) (0.004) 1949 to 1971 0.372 0.380 0.345 -0.035** (0.483) (0.485) (0.475) (0.004) 1949 to 1971 0.372 0.380 0.345 -0.034** (0.253) (0.248) (0.268) (0.002) 1972 to 1980 0.198 0.209 0.164 -0.044** (0.253) (0.248) (0.268) (0.002) Since 1991 0.093 0.071 0.163 0.092** (0.290) (0.257) (0.370) | | (2.790) | (2.903) | (2.060) | (0.025) |
| $\begin{array}{c ccccccccccccccccccccccccccccccccccc$ | Private Landlord | 0.789 | 0.789 | 0.789 | 0.000 |
| Subsidized Housing 0.110 0.131 0.047 -0.084^{**} (0.313) (0.338) (0.211) (0.003) Year of Construction 0.006 -0.011^{**} Before 1918 0.104 0.106 0.096 -0.011^{**} (0.305) (0.308) (0.294) (0.003) 1918 to 1948 0.165 0.168 0.154 -0.014^{**} (0.371) (0.374) (0.361) (0.004) 1949 to 1971 0.372 0.380 0.345 -0.035^{**} (0.483) (0.485) (0.475) (0.004) 1972 to 1980 0.198 0.209 0.164 -0.044^{**} (0.339) (0.406) (0.370) (0.002) Since 1990 0.069 0.066 0.078 0.012^{**} (0.290) (0.257) (0.370) (0.003) $2 = 500k$ (Center) 0.372 0.382 0.342 -0.041^{**} $\ge 500k$ (Suburb) 0.102 | | (0.408) | (0.408) | (0.408) | (0.004) |
| $\begin{array}{c c c c c c c c c c c c c c c c c c c $ | Subsidized Housing | 0.110 | 0.131 | 0.047 | -0.084^{***} |
| Year of Construction Before 1918 0.104 0.106 0.096 -0.011^{**} (0.305) (0.308) (0.294) (0.003) 1918 to 1948 0.165 0.168 0.154 -0.014^{**} (0.371) (0.374) (0.361) (0.004) 1949 to 1971 0.372 0.380 0.345 -0.035^{**} (0.483) (0.485) (0.475) (0.005) 1972 to 1980 0.198 0.209 0.164 -0.044^{**} (0.399) (0.406) (0.370) (0.004) 1981 to 1990 0.069 0.066 0.078 0.012* (0.253) (0.248) (0.268) (0.002) Since 1991 0.093 0.071 0.163 0.092** (0.290) (0.257) (0.370) (0.003) 2500k (Center) 0.372 0.382 0.342 -0.041^{**} (0.483) (0.486) (0.474) (0.004) ≥ 500k (Suburb) 0.102 0.111 0.077** | | (0.313) | (0.338) | (0.211) | (0.003) |
| Before 1918 0.104 0.106 0.096 −0.011** (0.305) (0.308) (0.294) (0.003) 1918 to 1948 0.165 0.168 0.154 −0.014** (0.371) (0.374) (0.361) (0.004) 1949 to 1971 0.372 0.380 0.345 −0.035** (0.483) (0.485) (0.475) (0.005) 1972 to 1980 0.198 0.209 0.164 −0.044** (0.399) (0.406) (0.370) (0.004) 1981 to 1990 0.069 0.066 0.078 0.012** (0.253) (0.248) (0.268) (0.002) Since 1991 0.093 0.071 0.163 0.092** (0.290) (0.257) (0.370) (0.003) 2 500k (Center) 0.372 0.382 0.342 −0.041** (0.483) (0.486) (0.474) (0.003) 100k to 500k (Center) 0.153 0.134 0.211 0.07** (0.360) | | Year of Co | onstruction | | |
| $\begin{array}{c ccccccccccccccccccccccccccccccccccc$ | Before 1918 | 0.104 | 0.106 | 0.096 | -0.011^{***} |
| $\begin{array}{cccccccccccccccccccccccccccccccccccc$ | | (0.305) | (0.308) | (0.294) | (0.003) |
| $\begin{array}{cccccccccccccccccccccccccccccccccccc$ | 1918 to 1948 | 0.165 | 0.168 | 0.154 | -0.014^{***} |
| $\begin{array}{c ccccccccccccccccccccccccccccccccccc$ | | (0.371) | (0.374) | (0.361) | (0.004) |
| $\begin{array}{c ccccccccccccccccccccccccccccccccccc$ | 1949 to 1971 | 0.372 | 0.380 | 0.345 | -0.035^{***} |
| $\begin{array}{c ccccccccccccccccccccccccccccccccccc$ | | (0.483) | (0.485) | (0.475) | (0.005) |
| $\begin{array}{c ccccccccccccccccccccccccccccccccccc$ | 1972 to 1980 | 0.198 | 0.209 | 0.164 | -0.044*** |
| $\begin{array}{c ccccccccccccccccccccccccccccccccccc$ | | (0.399) | (0.406) | (0.370) | (0.004) |
| $\begin{array}{c ccccccccccccccccccccccccccccccccccc$ | 1981 to 1990 | 0.069 | 0.066 | 0.078 | 0.012*** |
| $\begin{array}{c ccccccccccccccccccccccccccccccccccc$ | | (0.253) | (0.248) | (0.268) | (0.002) |
| $\begin{array}{c ccccccccccccccccccccccccccccccccccc$ | Since 1991 | 0.093 | 0.071 | 0.163 | 0.092*** |
| $\begin{array}{c c c c c c c c c c c c c c c c c c c $ | | (0.290) | (0.257) | (0.370) | (0.003) |
| $ \begin{array}{c ccccccccccccccccccccccccccccccccccc$ | City Size (Num | ber of Inha | bitants , $k = 10$ | 00 inhabitants | 3) |
| $\begin{array}{cccccccccccccccccccccccccccccccccccc$ | $\geq 500 \mathrm{k} \ \mathrm{(Center)}$ | 0.372 | 0.382 | 0.342 | -0.041*** |
| $ \geq 500k \text{ (Suburb)} \\ 0.102 \\ 0.111 \\ 0.076 \\ -0.034^{**} \\ (0.303) \\ (0.314) \\ (0.266) \\ (0.003) \\ (0.003) \\ 100k \text{ to } 500k \text{ (Center)} \\ 0.153 \\ (0.360) \\ (0.341) \\ (0.408) \\ (0.003) \\ (0.408) \\ (0.003) \\ (0.408) \\ (0.003) \\ (0.408) \\ (0.003) \\ (0.408) \\ (0.003) \\ (0.408) \\ (0.003) \\ (0.408) \\ (0.003) \\ (0.408) \\ (0.003) \\ (0.255) \\ (0.229) \\ (0.317) \\ (0.002) \\ (0.317) \\ (0.002) \\ (0.317) \\ (0.002) \\ (0.317) \\ (0.002) \\ (0.255) \\ (0.229) \\ (0.317) \\ (0.002) \\ (0.317) \\ (0.002) \\ (0.317) \\ (0.255) \\ (0.229) \\ (0.317) \\ (0.002) \\ (0.317) \\ (0.255) \\ (0.229) \\ (0.317) \\ (0.255) \\ (0.229) \\ (0.113) \\ (0.155) \\ (0.002) \\ (0.155) \\ (0.002) \\ (0.155) \\ (0.002) \\ (0.155) \\ (0.002) \\ (0.155) \\ (0.002) \\ (0.155) \\ (0.002) \\ (0.155) \\ (0.002) \\ (0.155) \\ (0.002) \\ (0.155) \\ (0.002) \\ (0.155) \\ (0.002) \\ (0.155) \\ (0.276) \\ (0.273) \\ (0.284) \\ (0.003) \\ (0.284) \\ (0.003) \\ (0.250) \\ (0.250) \\ (0.250) \\ (0.250) \\ (0.250) \\ (0.250) \\ (0$ | | (0.483) | (0.486) | (0.474) | (0.004) |
| $\begin{array}{cccccccccccccccccccccccccccccccccccc$ | $\geq 500 \mathrm{k} \; (\mathrm{Suburb})$ | 0.102 | 0.111 | 0.076 | -0.034^{***} |
| $\begin{array}{cccccccccccccccccccccccccccccccccccc$ | | (0.303) | (0.314) | (0.266) | (0.003) |
| $\begin{array}{cccccccccccccccccccccccccccccccccccc$ | 100k to $500k$ (Center) | 0.153 | 0.134 | 0.211 | 0.077*** |
| $\begin{array}{cccccccccccccccccccccccccccccccccccc$ | | (0.360) | (0.341) | (0.408) | (0.003) |
| $\begin{array}{cccccccccccccccccccccccccccccccccccc$ | 100k to 500k (Suburb) | 0.070 | 0.056 | 0.113 | 0.057*** |
| $\begin{array}{cccccccccccccccccccccccccccccccccccc$ | × , , | (0.255) | (0.229) | (0.317) | (0.002) |
| $\begin{array}{cccccccccccccccccccccccccccccccccccc$ | 50k to 100k (Center) | 0.030 | 0.032 | 0.025 | -0.007^{***} |
| $\begin{array}{cccccccccccccccccccccccccccccccccccc$ | | (0.171) | (0.176) | (0.155) | (0.002) |
| $\begin{array}{cccccccccccccccccccccccccccccccccccc$ | 50k to $100k$ (Suburb) | 0.026 | 0.019 | 0.048 | 0.028*** |
| $20k to 50k$ 0.083 0.081 0.089 0.008^{**} (0.276) (0.273) (0.284) (0.003) $5k to 20k$ 0.105 0.117 0.067 -0.050^{**} (0.306) (0.321) (0.250) (0.003) | × / | (0.161) | (0.138) | (0.213) | (0.001) |
| $\begin{array}{cccccccccccccccccccccccccccccccccccc$ | 20k to 50k | 0.083 | 0.081 | 0.089 | 0.008*** |
| 5k to 20k (0.105) (0.101) (0.003) (0.306) (0.321) (0.250) (0.003) | | (0.276) | (0.273) | (0.284) | (0.003) |
| (0.306) (0.321) (0.250) (0.003) | 5k to 20k | 0.105 | 0.117 | 0.067 | -0.050*** |
| | | (0.306) | (0.321) | (0.250) | (0.003) |

Table 1: Descriptive Statistics (Means and Standard Deviations/Standard Errors)

 $<\!\!{\rm continued}$ on next page>

| Variable | Full | Before Reform | After Reform | Difference |
|-----------------------|------------|---------------|--------------|----------------|
| < 5k | 0.058 | 0.068 | 0.029 | -0.039^{**} |
| | (0.235) | (0.252) | (0.169) | (0.002) |
| | Equij | oment | | |
| Central Heating | 0.924 | 0.912 | 0.963 | 0.051*** |
| | (0.264) | (0.284) | (0.190) | (0.002) |
| Balcony or Terrace | 0.675 | 0.659 | 0.725 | 0.066^{***} |
| | (0.468) | (0.474) | (0.446) | (0.004) |
| Basement | 0.919 | 0.919 | 0.919 | 0.000 |
| | (0.272) | (0.272) | (0.273) | (0.003) |
| Garden | 0.343 | 0.338 | 0.360 | 0.022*** |
| | (0.475) | (0.473) | (0.480) | (0.004) |
| | Further Ch | aracteristics | | |
| Rent-to-Income Ratio | 0.270 | 0.263 | 0.290 | 0.027*** |
| | (0.171) | (0.176) | (0.155) | (0.002) |
| Apt. Size per | 43.321 | 42.450 | 45.957 | 3.508^{***} |
| Person (sqm/person) | (21.242) | (21.077) | (21.520) | (0.194) |
| Rent Inexpensive | 0.336 | 0.351 | 0.287 | -0.065^{***} |
| (Assessment) | (0.472) | (0.477) | (0.452) | (0.005) |
| Rent Reasonable | 0.445 | 0.432 | 0.490 | 0.058*** |
| (Assessment) | (0.497) | (0.495) | (0.500) | (0.005) |
| Rent Expensive | 0.218 | 0.217 | 0.224 | 0.007 |
| (Assessment) | (0.413) | (0.412) | (0.417) | (0.004) |
| Apt. Size Too Small | 0.246 | 0.250 | 0.233 | -0.018^{***} |
| (Assessment) | (0.431) | (0.433) | (0.423) | (0.004) |
| Apt. Size Appropriate | 0.700 | 0.697 | 0.707 | 0.010** |
| (Assessment) | (0.458) | (0.459) | (0.455) | (0.004) |
| Apt. Size Too Large | 0.055 | 0.053 | 0.060 | 0.008*** |
| (Assessment) | (0.227) | (0.223) | (0.238) | (0.002) |
| Apt. Condition Good | 0.626 | 0.622 | 0.637 | 0.015*** |
| (Assessment) | (0.484) | (0.485) | (0.481) | (0.004) |
| Share of Observations | 100% | 75.2% | 24.8% | _ |

Table 1: Descriptive Statistics <continued>

Note: The samples are restricted to observations with elapsed tenure up to 10 years. Calculations use the SOEP sample weights. The table reports means (and standard deviations/standard errors in parentheses). The column labeled "Full" refers to the full sample. The columns labeled "Before Reform" and "After Reform" refer to the subsamples with tenancies starting before and after September 1, 2001, respectively. The column labeled "Difference" reports the mean difference between the two subsamples and its standard error. "Apt." denotes apartment, "Bldg." building, *,** and *** significance at the 10%-, 5%- and 1%-level, respectively. For sets of dummy variables the reference category is printed in bold. Variables reported under "Further Characteristics" are not used for the final regressions. Source: SOEP V28.1 and authors' calculations.

| Variable | Overall | 1st Tertile | 3rd Tertile | Difference |
|----------------------------|--------------|-------------|-------------|----------------|
| Rent (Deflated to 2005 €) | 429.786 | 245.422 | 651.916 | 406.494*** |
| | (208.231) | (54.982) | (201.564) | (1.433) |
| Apartment Size (in m^2) | 72.530 | 56.760 | 90.568 | 33.808*** |
| | (26.613) | (21.142) | (26.249) | (0.231) |
| Elapsed Tenure (in years) | 3.802 | 3.967 | 3.561 | -0.405^{***} |
| | (2.790) | (2.843) | (2.693) | (0.027) |
| Private Landlord | 0.789 | 0.707 | 0.873 | 0.165^{***} |
| | (0.408) | (0.455) | (0.333) | (0.005) |
| Subsidized Housing | 0.110 | 0.164 | 0.054 | -0.111^{***} |
| | (0.313) | (0.371) | (0.225) | (0.003) |
| Tenancy Starting | 0.248 | 0.198 | 0.283 | 0.085*** |
| After Reform | (0.432) | (0.398) | (0.450) | (0.004) |
| | Equipm | ent | | |
| Central Heating | 0.924 | 0.843 | 0.978 | 0.135*** |
| | (0.264) | (0.364) | (0.147) | (0.003) |
| Balcony or Terrace | 0.675 | 0.488 | 0.840 | 0.352^{***} |
| | (0.468) | (0.500) | (0.367) | (0.004) |
| Basement | 0.919 | 0.876 | 0.957 | 0.082^{***} |
| | (0.272) | (0.330) | (0.202) | (0.003) |
| Garden | 0.343 | 0.302 | 0.409 | 0.107^{***} |
| | (0.475) | (0.459) | (0.492) | (0.005) |
| Fi | urther Chara | acteristics | | |
| Rent-to-Income Ratio | 0.270 | 0.233 | 0.304 | 0.071*** |
| | (0.171) | (0.160) | (0.201) | (0.002) |
| Apt. Size per | 43.321 | 41.919 | 44.874 | 2.955*** |
| Person (sqm/person) | (21.242) | (19.584) | (23.328) | (0.209) |
| Rent Inexpensive | 0.336 | 0.477 | 0.200 | -0.276^{***} |
| (Assessment) | (0.472) | (0.499) | (0.400) | (0.005) |
| Rent Reasonable | 0.445 | 0.385 | 0.489 | 0.104*** |
| (Assessment) | (0.497) | (0.487) | (0.500) | (0.005) |
| Rent Expensive | 0.218 | 0.139 | 0.311 | 0.172^{***} |
| (Assessment) | (0.413) | (0.346) | (0.463) | (0.004) |
| Apt. Size: Too Small | 0.246 | 0.294 | 0.202 | -0.091^{***} |
| (Assessment) | (0.431) | (0.455) | (0.402) | (0.004) |
| Apt. Size: Appropriate | 0.700 | 0.679 | 0.710 | 0.032*** |
| (Assessment) | (0.458) | (0.467) | (0.454) | (0.004) |
| Apt. Size: Too Large | 0.055 | 0.028 | 0.087 | 0.059*** |
| (Assessment) | (0.227) | (0.165) | (0.282) | (0.002) |
| Good Apt. Condition | 0.626 | 0.555 | 0.691 | 0.136*** |
| (Assessment) | (0.484) | (0.497) | (0.462) | (0.005) |

Table 2: Further Descriptive Statistics: Means by Tertiles of Rents

Note: The samples are restricted to observations with elapsed tenure up to 10 years. Calculations use the SOEP sample weights. The table reports the overall mean, the mean in the first tertile (one third quantile), the mean in the third tertile (two third quantile) of the unconditional rent distribution, and the difference between the tertile-specific means. See Table 1 for further details.

| Variable | QR 17% | QR 50% | QR 83% | 83% - 17% | OLS |
|---|---------------------|---------------|--------------|---------------|----------------|
| Annua | l Reside | ncy Disc | ount | | |
| $0 \leq \text{Elapsed Tenure} \leq 3$ | -0.007^{*} | -0.014** | **-0.017** | * -0.010** | -0.012*** |
| | (0.004) | (0.002) | (0.003) | (0.004) | (0.002) |
| $4 \leq \text{Elapsed Tenure} \leq 10$ | -0.008^{**} | *-0.009** | *-0.009** | * -0.001 | -0.009^{***} |
| | (0.002) | (0.001) | (0.002) | (0.002) | (0.001) |
| | Reform | Effect | | | |
| After Reform | -0.020 | -0.029^{**} | -0.049** | * -0.029 | -0.041^{***} |
| | (0.018) | (0.012) | (0.014) | (0.020) | (0.012) |
| | Apartme | nt Size | | | |
| Ln(Size) | 0.498** | * 0.570** | * 0.558** | * 0.060 | 0.509*** |
| | (0.051) | (0.025) | (0.027) | (0.050) | (0.028) |
| $Ln(Size) \times Balcony$ | 0.098^{**} | * 0.103** | * 0.106** | * 0.007 | 0.119^{***} |
| | (0.025) | (0.020) | (0.020) | (0.028) | (0.018) |
| $Ln(Size) \times Basement$ | 0.198^{**} | * 0.152** | * 0.172** | * -0.026 | 0.175^{***} |
| | (0.049) | (0.029) | (0.029) | (0.051) | (0.028) |
| Ln(Size) (APE) | 0.741^{**} | * 0.775** | * 0.782** | * 0.041** | 0.745^{***} |
| | (0.016) | (0.011) | (0.012) | (0.018) | (0.011) |
| City Size by | ⁷ Popula | tion (Ref | Cat: < 5k | x) | |
| \geq 500k (Center) | 0.258** | * 0.262** | * 0.271** | * 0.013 | 0.263*** |
| | (0.023) | (0.018) | (0.018) | (0.026) | (0.017) |
| $\geq 500 \mathrm{k} (\mathrm{Suburb})$ | 0.208^{**} | * 0.202** | * 0.198** | * -0.010 | 0.200*** |
| | (0.024) | (0.018) | (0.020) | (0.026) | (0.018) |
| 100k to $500k$ (Center) | 0.155^{**} | * 0.134** | * 0.114** | * -0.041 | 0.127^{***} |
| | (0.023) | (0.018) | (0.019) | (0.026) | (0.017) |
| 100k to $500k$ (Suburb) | 0.108** | * 0.072** | * 0.049** | -0.059^{**} | 0.071*** |
| | (0.023) | (0.018) | (0.020) | (0.026) | (0.017) |
| 50k to $100k$ (Center) | 0.123** | * 0.089** | * 0.113** | * -0.010 | 0.085*** |
| | (0.030) | (0.023) | (0.023) | (0.033) | (0.022) |
| 50k to 100k (Suburb) | 0.059^{*} | 0.066** | * 0.038* | -0.021 | 0.060*** |
| × | (0.032) | (0.023) | (0.022) | (0.036) | (0.021) |
| 20k to 50k | 0.081** | * 0.048** | 0.032 | -0.049^{*} | 0.050*** |
| | (0.026) | (0.019) | (0.021) | (0.028) | (0.019) |
| 5k to 20k | 0.027 | 0.010 | 0.002 | -0.025 | 0.011 |
| | (0.025) | (0.019) | (0.021) | (0.028) | (0.018) |
| Missing Dummy | 0.287 | 0.275 | 0.142 | -0.144 | 0.188 |
| 0 | (0.477) | (0.171) | (0.154) | (0.497) | (0.175) |
| Type of House (Re | ef.cat.: De | etached H | [ouse, $1-2$ | Family) | |
| Other Bldg. | -0.039 | -0.002 | 0.021 | 0.060 | 0.000 |
| ~ | (0.061) | (0.039) | (0.033) | (0.062) | (0.031) |
| Nondetached House (1-2 Family) | 0.077** | * 0.092** | * 0.093** | * 0.017 | 0.088*** |

 Table 3: Baseline Rent Regressions

<continued on next page>

| Variable | QR 17% | QR 50% | QR 83% | 83% - 17% | OLS |
|-----------------------------|----------------|----------------|----------------|----------------|----------------|
| | (0.026) | (0.016) | (0.016) | (0.026) | (0.015) |
| Apt. in 3-4 Unit Bldg. | 0.097*** | 0.056*** | 0.035*** | * -0.062*** | 0.073*** |
| | (0.017) | (0.013) | (0.012) | (0.018) | (0.011) |
| Apt. in 5-8 Unit Bldg. | 0.125*** | 0.079*** | 0.066*** | * -0.060*** | 0.098*** |
| | (0.019) | (0.012) | (0.013) | (0.020) | (0.012) |
| Apt. in 9+ Unit Bldg. | 0.140*** | 0.111*** | 0.097*** | * -0.043** | 0.128*** |
| _ | (0.020) | (0.013) | (0.014) | (0.022) | (0.013) |
| High-Rise Apt. Bldg. | 0.131*** | 0.113*** | 0.097*** | * -0.034 | 0.122*** |
| | (0.036) | (0.024) | (0.027) | (0.038) | (0.025) |
| Missing Dummy | 0.018 | 0.026 | 0.080** | 0.063 | 0.046 |
| | (0.049) | (0.033) | (0.034) | (0.055) | (0.030) |
| Fu | rther Char | acteristi | cs | | |
| Private Landlord | 0.059*** | 0.075*** | 0.099*** | * 0.040*** | 0.078*** |
| | (0.011) | (0.008) | (0.009) | (0.012) | (0.008) |
| Private Landlord, Missing | 0.074^{***} | 0.058^{***} | 0.081*** | * 0.007 | 0.070*** |
| | (0.018) | (0.012) | (0.015) | (0.021) | (0.012) |
| Subsidized Housing | -0.097^{***} | -0.131^{***} | -0.159^{***} | -0.062^{***} | -0.132^{***} |
| | (0.011) | (0.009) | (0.010) | (0.013) | (0.008) |
| Subsidized Housing, Missing | 0.008 | 0.018 | 0.020 | 0.012 | 0.009 |
| | (0.044) | (0.024) | (0.030) | (0.046) | (0.025) |
| Selected Y | lear Dumm | ies (Ref. | cat.: 1984 | .) | |
| 2001 | 0.256*** | 0.225*** | 0.191*** | * -0.065** | 0.221*** |
| | (0.024) | (0.015) | (0.021) | (0.029) | (0.014) |
| 2011 | 0.198^{***} | 0.169^{***} | 0.171*** | * -0.027 | 0.179^{***} |
| | (0.025) | (0.020) | (0.025) | (0.033) | (0.018) |
| Year Dummies 1985-2011 | Х | х | х | | x |
| Year of Construction | Х | х | х | | х |
| State | х | х | х | | х |
| Equipment | х | х | х | | х |
| Residential Area | х | Х | Х | | х |
| Amenities | Х | Х | х | | х |

Table 3: Baseline Rent Regressions <continued>

Note: QR 17%, 50%, 83% denote quantile regressions at the three quantiles. 83% - 17% denotes the difference between the two quantiles 83% and 17%. Ref.cat. denotes the reference category. Calculations use the SOEP sample weights. Standard errors are in parentheses estimated by bootstrap with 200 replications, clustered at tenancy level. *,** and *** denote significance at the 10%-, 5%- and 1%-level, respectively. Average partial effects (APE) are partial effects at the mean of the covariates used for the interactions. Source: SOEP V28.1 and authors' calculations.

| | QR 17% | QR 50% | QR 83% | 83% - $17%$ | OLS |
|--|---------------|---------------|------------|-------------|----------------|
| | After Re | form Eff | fect | | |
| After Reform | -0.065** | *-0.071** | **-0.109** | * -0.044* | -0.096*** |
| | (0.022) | (0.014) | (0.016) | (0.024) | (0.014) |
| After Reform (APE) | -0.035^{**} | -0.043^{**} | **-0.066** | * -0.030 | -0.058^{***} |
| $(0 \leq \text{Elapsed Tenure} \leq 3)$ | (0.018) | (0.011) | (0.014) | (0.020) | (0.012) |
| After Reform (APE) | 0.032 | 0.006 | 0.001 | -0.031 | 0.013 |
| $(4 \leq \text{Elapsed Tenure} \leq 10)$ | (0.020) | (0.015) | (0.022) | (0.026) | (0.016) |
| Ann | ual Resi | dency D | iscount | | |
| | (a) Befo | ore Reform | n | | |
| $0 \leq \text{Elapsed Tenure} \leq 3$ | -0.014^{**} | *-0.020** | **-0.025** | * -0.012** | -0.021*** |
| | (0.004) | (0.003) | (0.004) | (0.005) | (0.003) |
| $4 \leq \text{Elapsed Tenure} \leq 10$ | -0.008^{**} | *-0.009** | **-0.009** | * -0.001 | -0.009^{***} |
| | (0.002) | (0.001) | (0.002) | (0.003) | (0.001) |
| | (b) Aft | er Reform | 1 | | |
| $0 \leq \text{Elapsed Tenure} \leq 3$ | 0.004 | -0.003 | 0.001 | -0.003 | 0.003 |
| | (0.006) | (0.005) | (0.005) | (0.007) | (0.004) |
| $4 \leq \text{Elapsed Tenure} \leq 10$ | 0.005 | -0.002 | 0.000 | -0.005 | 0.003 |
| | (0.004) | (0.004) | (0.005) | (0.006) | (0.004) |
| (c) Di | fference (A | fter minu | is Before) | | |
| $0 \leq \text{Elapsed Tenure} \leq 3$ | 0.018** | * 0.017** | ** 0.026** | * 0.009 | 0.023*** |
| | (0.007) | (0.005) | (0.006) | (0.008) | (0.005) |
| $4 \leq \text{Elapsed Tenure} \leq 10$ | 0.013** | * 0.007** | 0.009* | -0.004 | 0.012^{***} |
| | (0.005) | (0.004) | (0.005) | (0.006) | (0.004) |

Table 4: Annual Residency Discount Before and After Reform

Note: Pooled panel OLS and quantile regressions. Calculations use the SOEP sample weights. Standard errors are in parentheses and bootstrapped with 200 replications, clustered at tenancy level. *,** and *** denote significance at the 10%-, 5%- and 1%-level, respectively. Panel (a) shows annual discounts for tenancies where the old legal situation apply while panel (b) provides information on annual discounts for tenancies affected by the reform in 2001. Panel (c) shows the difference. Source: SOEP V28.1 and authors' calculations.

| | QR 17% | QR 50% | QR 83% | 83% - $17%$ | OLS |
|--|-------------|-------------|-----------|-------------|---------------|
| An | nual Resi | dency Dis | scount | | |
| | (a) Befo | ore Reform | | | |
| $0 \leq \text{Elapsed Tenure} \leq 3$ | -0.002 | -0.006*** | -0.006*** | * -0.004* | -0.004^{**} |
| | (0.002) | (0.002) | (0.002) | (0.002) | (0.002) |
| $4 \leq \text{Elapsed Tenure} \leq 10$ | 0.001 | 0.003*** | 0.005*** | * 0.004** | * 0.004*** |
| | (0.001) | (0.001) | (0.001) | (0.001) | (0.001) |
| | (b) Aft | er Reform | | | |
| $0 \leq \text{Elapsed Tenure} \leq 3$ | 0.006 | 0.003 | 0.004 | -0.003 | 0.007** |
| | (0.004) | (0.003) | (0.004) | (0.003) | (0.003) |
| $4 \leq \text{Elapsed Tenure} \leq 10$ | 0.005 | 0.008*** | 0.010*** | * 0.005 | 0.007^{**} |
| - · - | (0.003) | (0.002) | (0.003) | (0.003) | (0.003) |
| (c) Di | fference (A | fter minus | Before) | | |
| $0 \leq \text{Elapsed Tenure} \leq 3$ | 0.008^{*} | 0.009** | 0.010** | 0.001 | 0.011*** |
| | (0.005) | (0.004) | (0.004) | (0.003) | (0.004) |
| 4 < Elapsed Tenure < 10 | 0.004 | 0.005^{*} | 0.004 | 0.000 | 0.003 |
| - • - | (0.004) | (0.003) | (0.003) | (0.003) | (0.003) |

Table 5: Annual Residency Discount Before and After Reform (Tenancy Fixed Effects)

Note: Panel OLS and quantile regressions. The estimates account for tenancy mean fixed effects. Calculations use the SOEP sample weights. Standard errors are in parentheses and bootstrapped with 200 replications, clustered at tenancy level. *,** and *** denote significance at the 10%-, 5%- and 1%-level, respectively. Panel (a) shows annual discounts for tenancies where the old legal situation apply while panel (b) provides information on annual discounts for tenancies affected by the reform in 2001. Panel (c) shows the difference. Source: SOEP V28.1 and authors' calculations.